Dear Editors:

We have revised our manuscript in light of your comments and enclosed herewith for your consideration is the third revised version of our manuscript (MS:6922772449127051) entitled “Identification of genes associated with cisplatin resistance in human oral squamous cell carcinoma cell line” by Ping Zhang, Zhiyuan Zhang, Xiaojian Zhou, Weiliu Qiu, Fangan Chen and Wantao Chen.

We appreciate the point, the English language use through the paper remains poor in places, made by editors and we are in general in agreement that you recommended would strengthen our paper. Accordingly, we have asked 'Manuscript Presentation Service' based at the University of Aberdeen to help us copyedit the paper.

We hope that our manuscript is now acceptable for publication as early as it could, but please let me known if we can provide further information.

I look forward to hearing from you soon.

Sincerely yours,

Wantao Chen
Ninth People’s Hospital
Shanghai JiaoTong University
639,Zhizaoju Road
Shanghai, 200011
China
Tel: 86-21-63138341-5211
Fax: 86-21-63135412
Email: chenwantao2002@hotmail.com
June 18,2006